Author’s response to reviews

Title: Linkages between health and agriculture sectors in Ethiopia: a formative research study exploring barriers, facilitators and opportunities for local level coordination to deliver nutritional programmes and services

Authors:
Girmay Mersha (girmayayana@yahoo.com)
Tesoaye Bekele (tesfayehai@yahoo.com)
Desalegn Goa (desukuche@gmail.com)
Andinet Hailu (andinet_abera@yahoo.com)
Solomon Hailu (solomoneshetu13@yahoo.com)
Alemnesh Amen (amenalem2005@gmail.com)
Aweke Arfasa (awekekeb@gmail.com)
Masresha Anegago (masresha88@gmail.com)
Cami Allen (cami.allen@lshtm.ac.uk)
Mihretab Melese (mihretab.salasibew@lshtm.ac.uk)
Dangour Alan (alan.dangour@lshtm.ac.uk)

Version: 1 Date: 26 May 2017

Author’s response to reviews:

Date: 26 May 2017

TO: The Editor

BMC Nutrition

Dear Sir or Madam:

We would like to take this opportunity to thank you for considering our manuscript for publication in your journal and we appreciate your comments and suggestions which certainly improved our manuscript entitled “Linkages between Health and Agriculture sectors in Ethiopia:
a formative research study exploring barriers, facilitators and opportunities for local level coordination in delivering nutritional programmes and services”.

As requested, we have addressed all comments made by the editor and reviewers, revised the manuscript and highlighted the changes we made in yellow, and provided you with point by point response for the comments made.

We believe, and as acknowledged by the reviewers, our manuscript contributes to the existing knowledge gap by identifying key barriers, facilitators and opportunities for effective multi-sectoral coordination in order to provide nutritional services to the community. Therefore, we kindly request you to progress our manuscript towards publication in your journal.

With best regards

Girmay Ayana

Corresponding Author